\documentclass[12pt]{article}
\usepackage{geometry}
\paperwidth 534pt
\usepackage[utf8]{inputenc}
\usepackage[T2A]{fontenc}
\usepackage[russian,english]{babel}
\usepackage{graphicx}
\usepackage{wrapfig}
\usepackage{hyperref} 
\usepackage{rotating}
\usepackage[export]{adjustbox}
\topmargin -88pt
\textheight 750pt
%\oddsidemargin -20pt
\oddsidemargin -60pt
\textwidth 512pt
%\title{Chulym arsenal (Чулымский арсенал)~ ~ ~ ~ ~ ~ ~ }
%\author{Чулымский а ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~}
\usepackage[yyyymmdd]{datetime}
\renewcommand{\dateseparator}{.}
\newcommand \ing \includegraphics
\newcommand \sx \scalebox

\setlength{\footskip}{16pt}

\makeatletter
\let\latexl@section\l@section
\def\l@section#1#2{\begingroup\let\numberline\@gobble\latexl@section{#1}{#2}\endgroup}
\makeatother

\begin{document}
\begin{thebibliography}{11} \label{references}\vspace{-6pt}
%\begin{spacing}{0.9}
\addcontentsline{toc}{section}{References}
%\input{HeidiRref.tex}

%\bibitem{adn}
\small
\setlength{\itemsep}{1pt}
\setlength{\parskip}{1pt}
\baselineskip 6pt

\bibitem{adn}
\url{https://web.archive.org/web/20170724022955/http://www.rusprofile.ru/id/650223}
Полное наименование (2023)
ФЕДЕРАЛЬНОЕ ГОСУДАРСТВЕННОЕ КАЗЕННОЕ УЧРЕЖДЕНИЕ КОМБИНАТ "ТЕХНИКА" УПРАВЛЕНИЯ ФЕДЕРАЛЬНОГО АГЕНТСТВА ПО ГОСУДАРСТВЕННЫМ РЕЗЕРВАМ ПО СИБИРСКОМУ ФЕДЕРАЛЬНОМУ ОКРУГУ Регион Новосибирская область Юридический адрес 632553, Новосибирская область, Чулымский район, город Чулым-3, Октябрьская улица, 1 ..
Руководитель Директор
Шабанов Александр Николаевич
с 26 сентября 2018 г.

\bibitem{coo}
\url{https://satellites.pro/Russia_map#55.0951,80.7841,18}

\bibitem{vk3287}
\url{https://vk.com/club6028488} Армия.Чулым-3.в/ч 3287. 290 members (2023)

\bibitem{mapia}
\url{http://wikimapia.org/index_wjsl.php#lang=en&lat=55.109113&lon=80.797234&z=13&m=w}

\bibitem{z}
\url{https://www.youtube.com/watch?v=xaQ2rdBO6-I&t=12s}
ПРИВЕТ СЫНОЧКА! Как ты котик? КАК ТВОЯ СПИНКА? СКУЧАЕМ/ ``ПУСЯ, Я ТОЖЕ СКУЧАЮ СИЛЬНО''/ @VolodymyrZolkin
Volodymyr Zolkin//
Mar 5, 2023//
Крикунов Петр Васильевич
\_\_\_
00:00 Начало (за кадром)
01:53 Представтесь 
04:06 Образование. Работа
09:56 После театра пошел работать охранником
12:48 После охранником золота пошел работать электромонтером
13:35 Как вас мобилизовали?
16:32 Когда вы зашли на территорию Украины ?
18:10 Когда попали "в лапы ВСУ"?
21:15 Как вам робота наших ВСУ?
24:38 Опишите свой последний день на свободе
30:04 Уровень наших военных как вам?
32:26 КАКАЯ ВАША ЦЕЛЬ ЗДЕСЬ?
36:01 Звонок на россию матери
38:33 Звонок на россию жене
44:13 Кому нужна эта война? Кто победит?
46:14 Привет на россию.
- - - 
Крикунов Петр Васильевич (Krikunov Petr Vasilievich, 1989.01.31, Камчатка).. 
Чулым 180 км от Томска. ..
16 складов .. Периметр 8km.

%\bibitem{sha}
%\url{https://www.rusprofile.ru/id/650223&ved=2ahUKEwjCu7We5d39AhULlFYBHQ_TCKAQFnoECAIQAw&usg=AOvVaw1i_9ih_VUOm_5f8KNsJqXk}
%\url{https://www.rusprofile.ru/id/650223&ved=2ahUKEwjCu7We5d39AhULlFYBHQ_TCKAQFnoECAIQAw}
%ФГКУ Комбинат "Техника" Росрезерва Чулымский район, городок Чулым-3, Октябрьская ул., д. 1 зарегистрирована 26.11.2002 регистратором МЕЖРАЙОННАЯ ИНСПЕКЦИЯ ФЕДЕРАЛЬНОЙ НАЛОГОВОЙ СЛУЖБЫ № 16 ПО НОВОСИБИРСКОЙ ОБЛАСТИ. Руководитель организации: директор Шабанов Александр Николаевич.
%(2002-2023)
%01
%\url{http://www.fh-augsburg.de/~harsch/germanica/Chronologie/19Jh/Spyri/spy_intr.html}
%Johanna Spyri  1827 - 1901 // Zum 175. Geburtstag  am 12. Juni 2002. 
%Johanna Spyri wird 1827 im Hirzel, Kanton Zürich, als Johanna Louise Heusser geboren, das vierte von sechs Kindern. ..

\bibitem{the}
\url{https://mizugadro.mydns.jp/t/index.php/New_World_Order}

\end{thebibliography}

\parindent 0pt
\parskip 4pt
\renewcommand\thesection{}
\renewcommand{\figurename}{Fig.}
\addto\captionsenglish{\renewcommand{\figurename}{Fig.}}
%\maketitle

%\begin{wrapfigure}{R}{200px}
%\beginparbox{312px}
\parbox{280px}{
\vskip -18pt
{\LARGE \bf Chulym аrsenal} 

{\Large \bf (Чулымский арсенал)} 

\large
\vskip 8pt
\baselineskip 15pt
\parskip 4pt
Chulym arsenal (Chulymski arsenal)\\
%, Чулымский арсенал) 
is big Russian military warehouse\\
 at Novosybirsk Oblas.
 %t, Russia.
Official name\cite{adn}:

Федеральное государственное казенное\\ учреждение комбинат ``Техника''\\
Управления Федерального агентства по\\
государственным резервам по Сибирскому\\
федеральному округу

The same object is denoted also with term «в/ч 3287» («Vh3287»). For year 2023, аmount of personal of «Vh3287» is estimated to be of order of 290.\cite{vk3287}

Chulym arsenal is suspected to supply the Russian army during the putin world war and, especially,
at the Russian invasion into Ukraine.\\
Publications about this arsenal could be useful in construction of historic models.
\vspace{-18pt}

\section{01. Location}

\vspace{-8pt}
Post address \cite{adn}:
Новосибирская обл., 
%Чулымский район, 
город Чулым-3, Октябрьская ул., д.1

 Coordinates:\\
 55.0905,80.9785 office (that Gogle suggests)\\
 55.1099,80.7966 Chulym-3\\ 
 55.0949,80.7837 (railroad station \cite{coo})\\
55.0949,80.79 Main gate
\vskip -32pt

~

\section{02. Access}
\vskip -8pt

Both railroad access and the car road access\\ to Chulym arsenal and Chulym-3 are shown\\
by Wikimapia \cite{mapia}; enlarged notes are added:

\vskip 190px

\sx{.25}%{\begin{picture}(2,052,801)
{\begin{picture}(2,052,380)
\put(-10,-20){\ing{Chulym3mapiaC.jpg}}
\put(40,780){\sx{3}{\sf Kokoshino}}
\put(-14,680){\sx{3}{\sf Kokoshino}}
\put(4,648){\sx{3}{\sf railroad}}
\put(14,616){\sx{3}{\sf station}}
\put(1020,480){\sx{3}{\begin{rotate}{-13}{\sf Railroad Omsk - Novosibirsk}\end{rotate}}}
\put(850,404){\sx{3}{\sf Chulym-3}}
\put(1850,200){\sx{3.8}{\sf Chulym}}
\put(1090,360){\sx{3}{\sf Vh3287}}
\put(450,190){\sx{3}{\sf Chulym arsenal railroad station}}
\put(1030,220){\sx{3}{\sf main gate}}
\put(740,20){\sx{3}{\sf Chulym arsenal}}
\end{picture}}

} ~
\parbox{210px} 
{
\vskip -14pt
\ing[width=220px]{ChulymMap.jpg}\vskip 2pt

\ing[width=220px]{ChulymArsenalLeninaKirovaMap.jpg}
\vspace{-20pt}

\sx{.85}[.9]{~Чулымский р-н, Чулым-3, Октябрьская ул.,д.1}
\vskip 4pt

\ing[width=220px]{Chulym3map.jpg}%\vskip -26pt

\vspace{-18pt}

\center{\small Чулым-3 and Chulym arsenal}\vskip 6pt

\ing[width=220px]{ChulymArsenalSatFragme.jpg}
\vskip -4pt
%\caption{\label{fig:frag}
{\small North-East part: 55.0949,80.7837}
\vskip 180pt
%\end{wrapfigure}
}

\newpage
\section{03. Personal}

\begin{wrapfigure}{R}{80pt}
\vskip -48pt
\ing[width=80pt]{KrikunovPetrVasilievich01.jpg}
\vskip -4pt

~~Krikunov P.\cite{z}
\vskip -48pt

\end{wrapfigure}
\vskip -8pt
\baselineskip 9pt
\parskip 2pt

Крикунов Петр Васильевич (Krikunov Petr Vasilievich, 1989.01.31, Камчатка), vigilant, 2020 \cite{z}

Шабанов Александр Николаевич (Shabanov Aleksandr Nikolaevich), director (2023) \cite{adn}.

\vspace{-18pt}

\section{04. Russian invasion into Ukraine}
\vskip -8pt

During the Russian invasion into Ukraine, Krikunov Petr Vasilievich works at the Chulym arsenal.

Then, during the full-scale invasion (спецоперация) Krikunov performs the aggressive war
and got captured as war prisoner. 

The Russian troops behave as barbatian, orks, that attack mainly civilian objevts.

Such a style of aggression is qualified as a war crime.

The transfer of hostilities to the aggressor territory is expected. This seems to be the only way to stop the Russian aggression \cite{the}

Then, the Russian military objects, even at territory of Russia, can be treated as legitimate targets.

One may expect some bavovna to happen at all Russian military objects,
especially at the war industry and arsenals auch as the Chulym arsenal.

\vspace{-18pt}

\section{05. Warning}

\vskip -8pt


Publications about the Chulym аrsenal are collected and analysed at Mizugadro with scientific goals.

Venganza does not belong to the set of these goals.

The consideration above is not a suggestion for the place of a bavovna
at the transfer of hostilities to aggressor territory.

\vspace{-8pt}

\makeatletter 
\renewcommand\@biblabel[1]{\hspace{-8pt}#1.} 
\makeatother
\renewcommand{\refname}{References}  % for the article class
\renewcommand{\bibname}{References}  %for book

\baselineskip 9pt
\parskip 0pt

  
 \normalsize
 \baselineskip 0pt
\parskip 0pt

\vspace{-18pt}

\section{Keywords}
\vspace{-8pt}\small

\url{https://mizugadro.mydns.jp/t/index.php/Bavovna}

\url{https://mizugadro.mydns.jp/t/index.php/Chulym_arsenal}

\url{https://mizugadro.mydns.jp/t/index.php/Designate_Russia_as_state_sponsor_of_terrorism}

\url{https://mizugadro.mydns.jp/t/index.php/I_bombed_Dresden}

\url{https://mizugadro.mydns.jp/t/index.php/Putin_world_war}

\url{https://mizugadro.mydns.jp/t/index.php/Russian_invasion_into_Ukraine}

\url{https://mizugadro.mydns.jp/t/index.php/New_World_Order}

\url{https://mizugadro.mydns.jp/t/index.php/Transfer_of_hostilities_to_aggressor_territory}


%\href{https://mizugadro.mydns.jp/t/index.php/Агрессор_и_оккупант}{test}
%{https://mizugadro.mydns.jp/t/index.php/Агрессор_и_оккупант}
%\url{https://mizugadro.mydns.jp/t/index.php/Бавовна}
%\url{https://mizugadro.mydns.jp/t/index.php/Большевики_убили_почти_всех}
%\url{https://mizugadro.mydns.jp/t/index.php/Перенос_боевых_действий_на_территорию_страны_агрессора}
%\url{https://mizugadro.mydns.jp/t/index.php/Спецоперация}
%\url{https://mizugadro.mydns.jp/t/index.php/Ст.353_УК_РФ}


%\renewcommand{\contentsname}{Contents:}
%\tableofcontents


\end{document}
